A Latent Markov Modelling Approach to the Evaluation of Circulating Cathodic Antigen Strips for Schistosomiasis Diagnosis Pre- and Post-Praziquantel Treatment in Uganda

Artemis Koukounari^{1,2}*, Christl A. Donnelly¹, Irini Moustaki², Edridah M. Tukahebwa³, Narcis B. Kabatereine³, Shona Wilson⁴, Joanne P. Webster⁵, André M. Deelder⁶, Birgitte J. Vennervald⁷, Govert J. van Dam⁶

1 MRC Centre for Outbreak Analysis and Modelling, Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom, 2 Department of Statistics, London School of Economics and Political Science, London, United Kingdom, 3 Schistosomiasis Control Initiative at Vector Control Division – Ministry of Health, Kampala, Uganda, 4 Department of Pathology, University of Cambridge, Cambridge, United Kingdom, 5 Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom, 6 Department of Parasitology, Leiden University Medical Center, Leiden, The Netherlands, 7 Section for Parasitology and Aquatic Diseases, University of Copenhagen, Denmark

Abstract

Regular treatment with praziguantel (PZQ) is the strategy for human schistosomiasis control aiming to prevent morbidity in later life. With the recent resolution on schistosomiasis elimination by the 65th World Health Assembly, appropriate diagnostic tools to inform interventions are keys to their success. We present a discrete Markov chains modelling framework that deals with the longitudinal study design and the measurement error in the diagnostic methods under study. A longitudinal detailed dataset from Uganda, in which one or two doses of PZQ treatment were provided, was analyzed through Latent Markov Models (LMMs). The aim was to evaluate the diagnostic accuracy of Circulating Cathodic Antigen (CCA) and of double Kato-Katz (KK) faecal slides over three consecutive days for Schistosoma mansoni infection simultaneously by age group at baseline and at two follow-up times post treatment. Diagnostic test sensitivities and specificities and the true underlying infection prevalence over time as well as the probabilities of transitions between infected and uninfected states are provided. The estimated transition probability matrices provide parsimonious yet important insights into the re-infection and cure rates in the two age groups. We show that the CCA diagnostic performance remained constant after PZQ treatment and that this test was overall more sensitive but less specific than single-day double KK for the diagnosis of S. mansoni infection. The probability of clearing infection from baseline to 9 weeks was higher among those who received two PZQ doses compared to one PZQ dose for both age groups, with much higher re-infection rates among children compared to adolescents and adults. We recommend LMMs as a useful methodology for monitoring and evaluation and treatment decision research as well as CCA for mapping surveys of S. mansoni infection, although additional diagnostic tools should be incorporated in schistosomiasis elimination programs.

Citation: Koukounari A, Donnelly CA, Moustaki I, Tukahebwa EM, Kabatereine NB, et al. (2013) A Latent Markov Modelling Approach to the Evaluation of Circulating Cathodic Antigen Strips for Schistosomiasis Diagnosis Pre- and Post-Praziquantel Treatment in Uganda. PLoS Comput Biol 9(12): e1003402. doi:10.1371/journal.pcbi.1003402

Editor: Mark M. Tanaka, University of New South Wales, Australia

Received May 6, 2013; Accepted October 28, 2013; Published December 19, 2013

Copyright: © 2013 Koukounari et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The work presented in this article was supported through a Population Health Scientist Fellowship awarded to AK jointly funded by the Medical Research Council (MRC) and the Department for International Development (DFID) under the MRC/DFID Concordat agreement (G0902130/1). CAD thanks the MRC for Centre funding. The field work and data collection were funded by the Commission of the European Community's Science and Technology for Development Programme (INCO-DEV Contract No: 517733 MUSTSchistUKEMA). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: artemis.koukounari@kcl.ac.uk

Introduction

Schistosomiasis is a debilitating parasitic disease in tropical and sub-tropical areas which has recently attracted increased focus and funding for control through large scale mass drug administration (MDA) of praziquantel (PZQ) [1]. However the ability of current control initiatives to operate cost effectively is reduced by technical limitations of currently available schistosomiasis diagnostics [2,3]. With the recent resolution on schistosomiasis elimination by the 65th World Health Assembly, schistosomiasis diagnostics research for population-based assessment is critical with careful consideration given to how those tools might be used within disease elimination programmes [4–6].

At present the World Health Organization (WHO) recommends the Kato-Katz (KK) method as the standard tool for the qualitative and quantitative diagnosis of *Schistosoma mansoni* infection because of its assumed high specificity, relative simplicity in field conditions and attractive price. WHO also recommends MDA with PZQ to be delivered to community populations defined where KK surveys show an estimated prevalence of over

Author Summary

Schistosomiasis remains one of the most prevalent parasitic diseases in developing countries, with Schistosoma mansoni being the most widespread of the humaninfecting schistosomes. For the routine surveillance of human S. mansoni infection more "field-applicable," sensitive, and cost-effective diagnostics that replicate faecal samples over several consecutive days [the Kato-Katz (KK) method], are needed. We propose a statistical modelling framework in order to evaluate the diagnostic performance of the urine strip test for Circulating Cathodic Antigen (CCA) and single-day double KK measurements over three consecutive days for the diagnosis of S. mansoni infection in two different age groups from Uganda preand post- praziguantel (PZQ) treatment. We demonstrate that CCA is an appropriate tool for mapping surveys of S. mansoni infection. Our findings should allow for evaluation of the risk of potential misinterpretation with regards to diagnosis of S. mansoni infection through CCA or KK in this endemic setting pre- and post- PZQ treatment as the numbers and infection intensities are brought down, bridging existing important gaps in schistosomiasis diagnostics research. More generally, the proposed statistical analysis can reveal important biological insights from other diseases without gold standard diagnostic tools whenever longitudinal data are available.

50% in school-aged children and to be delivered to children aged 6-16 years where the estimated prevalence is between 10% and 50% in this age group. However, it is well known that KK method from single stool samples, particularly at low infection endemicities and following PZQ MDA, can underestimate Schistosoma infection prevalence (and intensities) and thus overestimate cure rates [7-11], whilst even multiple slides over multiple days of stool sampling can influence specificity and overestimate prevalence [10–15]. The necessity for more "field-applicable", sensitive and cost-effective diagnostics than the KK method, at least for the routine surveillance of human S. mansoni infection such as that inherent within mapping of at-risk populations has also been recently highlighted [16,17]. Even more worrying, in some endemic regions microscopic stool samples examination is considered too logistically difficult in terms of personnel available for routine surveillance and therefore S. mansoni infection remains undetected and untreated in control programmes [18]. A promising diagnostic option is a urine strip test for Circulating Cathodic Antigen (CCA) which is a genus-specific glycan regurgitated by adult schistosome worms into the blood stream [19]. A number of cross-sectional studies evaluating CCA accuracy pre-treatment have stressed the need for further research assessing the potential role of this diagnostic assay at different stages of schistosomiasis control programs [15,17,20].

In this study, we assessed for the first time the diagnostic accuracy of CCA and of double KK faecal slides over three consecutive days for *S. mansoni* infection by age group at baseline and at two follow-up times post treatment which should be viewed as a proxy for low transmission areas. Because no true gold standard diagnoses are available, we developed and fitted latent Markov models (LMMs) to estimate diagnostic test sensitivities and specificities, the true underlying infection prevalence and the probabilities of transitions between infected and uninfected states [21]. LMMs - which are sometimes also referred to as latent transition models or regime-switching models – are used to analyze discrete-time longitudinal data where respondent observations contain measurement error. This approach defines the

true states as categories (latent classes) of a dynamic latent (unobservable) variable within a statistical model. The Markov assumption is reflected in the model via transition probabilities which allow for correlation between a respondent's true state at times t-1 and t [22].

We analyzed a detailed dataset from a longitudinal cohort living along the shorelines of Lake Victoria in Uganda who received one or two doses of PZQ treatment at baseline. We demonstrated how the use of LMMs allows estimation of the 'true' prevalence of *S. mansoni* infection over time and the quantification of the additional benefit of a second PZQ dose in reducing re-infection levels by age group.

Materials and Methods

Ethics Statement

Ethical clearance was obtained from the Uganda National Council of Science and Technology and the study was also presented to the Danish National Committee on Biomedical Research Ethics in Denmark (Reference Number: UNCST: HS 59). Informed consent was obtained from individual adult participants but for children the parents or guardians consented on their behalf. Thereafter, each individual signed a consent form before any activity started. All information obtained from participants was kept confidential. Because some of the participants might have potentially received PZQ treatment recently through MDA, field survey assistants asked each participant detailed questions about previous treatment in order to exclude such individuals ever treated through MDA, although no such pretreated individuals were identified in the current study.

Study Design and Population

We conducted our study in Musoli village, Mayuge district at baseline and nine weeks after treatment during 2005. Participants of this study were randomly allocated to one or two PZQ (Shin Poong Pharmaceuticals, Seoul Republic of Korea) doses at baseline (40 mg PZQ per kg body weight; double treatment group: two times PZQ 40 mg per kg body weight administered two weeks apart). The field survey assistants who delivered these treatments were not aware of the infection status of any participants at any time. First follow-up data collection was performed at nine weeks, and hence at a time aimed to assess cure rates where the risk of any eggs detected occurring as a consequence of reinfection was minimal. Second follow-up data collection was performed two years later during 2007. With regards to treatment after two years, this is part of the National programme to treat everyone living in an endemic area and thus a second MDA was offered after two years. The study location was selected as this is an area of Uganda known for perennial S. mansoni transmission, situated on the shore of Lake Victoria [23] where the community population is not targeted by the National Control Program [24,25]. The community consists primarily of fishermen and their dependants with the lake being the only source of fresh water for them. Furthermore, infrequent use and/or availability of latrines leads to contamination of the lake especially near the shoreline, where there is underwater vegetation suitable for the aquatic intermediate host snails to thrive. In addition, only individuals >6 years of age were enrolled as very often this is the age from which schistosome-induced morbidity, in general, becomes evident.

Population Census

Before any data collection took place, trained and experienced demographers conducted in 2005 a census of the village

population. During this process the inhabitants in each household registered their relation to the head of the household, year of birth, gender, occupation, duration of residency in the village and tribal membership.

Sample Size

A stratified random sample for age and sex was then selected from the census data. Sample size calculations included detection of a significant difference of cure rates between the two treatment groups with reference to cure rates obtained along Lake Albert. A significance level of $\alpha = 0.05$, a power of 90% with a drop-out rate of 40% over the two years of studies contributed to the calculation of 552 individuals but at baseline we managed to recruit 446 with full parasitological and CCA in urine data as described elsewhere [23,26]. A further six cases were then eliminated from the analysis presented in this study due to missing data in number of treatments and their age.

Parasitological and Immunological Diagnosis

Stool samples were collected on three consecutive days from each member of the cohort and examined for the presence of *S. mansoni* ova. Two duplicate slides of each stool sample were examined using the KK technique at each day. Each slide was read by two trained microscopists and any discrepancies resolved before results were recorded as eggs per gram (EPG) faeces. The results for all six slides were combined for the descriptive results presented in Table 1 as this is considered the best KK diagnostic performance scenario [13] while results of two slides of each single day are combined and incorporated in the statistical models' derived results (Tables 2A–3B) because in many studies only a single stool sample is analysed [25,27–29].

Single urine samples were kept cold after collection and after return to the laboratory at the end of the day aliquoted and stored frozen. At baseline and nine weeks, CCA urine samples were kept in a freezer in Uganda for nine months before they were transferred frozen to Leiden. At the two year follow-up CCA urine samples were delivered to Leiden within four weeks of collection. In Uganda, all CCA urine samples were kept at -20 Celsius degrees. The freezer in which urine was stored was -18 to -20degrees C. Results should not be affected as the antigen is very stable and its detection is not influenced by periods of being frozen, freeze thaw cycles, or even storage for weeks at room temperature. They were kept frozen during transport to the Department of Parasitology, in Leiden, The Netherlands, where the CCA urine assays were performed as previously described [19,30]. Briefly, for the laboratory-based test, 25 µL of completely thawed and vortexed urine was added to a tube containing dried carbon conjugated antibody, along with 75 µL of buffer and mixed well. Test strips were added, and allowed to develop for 40 minutes. Strips were removed, allowed to dry, and read against a set of five Quality Control (QC) standards of 0, 10, 100, 1000 and 10 000 ng of semi-purified worm antigen (containing CCA) per ml negative urine. A score of 0 indicated a negative result; the 0.5 stands for trace, while 1, 2, and 3, indicated that the intensity of the test line was similar to that of the respective 100, 1000, and

Table 1. 2×2 descriptive tables for *S. mansoni* infection as determined by CCA and microscopy over time; sensitivity and specificity of CCA were estimated assuming that the combination of 6 KK measurements over 3 days is 100% sensitive and 100% specific.

A: for children (ag	e 7–16)						
	Baseline		9 weeks	9 weeks		2 years	
	KK (6 measurements over 3 days)						
ССА	-	+	-	+	-	+	
-	2	17	39	26	4	7	
+	2	148	17	66	7	101	
Empirical CCA diagr	nostic performant	ce (%) assuming KK (6	measurements over 3 day.	s) as "the gold stand	lard″		
Sensitivityª	89.7 (84.0 to 93.9) ^c		71.7 (61.4 to	71.7 (61.4 to 80.6)		93.5 (87.1 to 97.4)	
Specificity ^b	50.0 (6.8 to 93.2)		69.6 (55.9 to	69.6 (55.9 to 81.2)		36.4 (10.9 to 69.2)	
B: for adolescents	and adults (age	e 17–76)					
	Baseli	ne	9 weeks	9 weeks		2 years	
	KK (6 measurements over 3 days)						
ССА	_	+	-	+	_	+	
-	37	41	124	24	54	29	
+	7	189	32	43	30	80	
Empirical CCA diagn	nostic performant	ce (%) assuming KK (6	measurements over 3 day.	s) as "the gold stand	ard"		
Sensitivityª	82.2 (7	6.6 to 86.9)	64.2 (51.5 to	64.2 (51.5 to 75.5)		73.4 (64.1 to 81.4)	
Specificity ^b	84.1 (69.9 to 93.4)		79.5 (72.3 to	79.5 (72.3 to 85.5)		64.3 (53.1 to 74.5)	

^aThis is derived by taking the ratio of "true positives" over the sum of "true positives" and "false negatives". For instance for children at baseline number of "true positives" is 148 and number of "false negatives" is 17.

^bThis is derived by taking the ratio of "true negatives" over the sum of "true negatives" and "false positives". For instance for children at baseline number of "true negatives" is 2 and number of "false positives" is 2.

^c95% Exact binomial confidence intervals are provided in brackets.

General note: Complete case data at each time point are displayed and contributed to the calculations of empirical sensitivities and specificities.

CCA KK (2 measurements) Sensitivity Sensititity Sensitivity Sen	tts) 1 st day KK (2 measurements) 2 nd day KK (2 measurement Specificity Sensitivity Specificity Sensitivity
Sensitivity Sensitivity Sensitivity Sensitivity Baseline 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 82.0 (56.9 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 89.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 8 for adolescents and adults-n= 273 Attion of the second seco	Specificity Sensitivity Specificity Sensitivity
Baseline 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (56.9 to 100.0) 8 2 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 2 years 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 8: for adolescents and adults-n= 273 8 8 8 8 CCA CCA 8 8 8 8	
9 weeks 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 82.0 (56.9 to 100.0) 8 2 years 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 2 years 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 B: for adolescents and adults-n= 273 CCA Sensitivity Specificity Specificity Sensitivity	71.5 (53.4 to 84.6) 93.8 (88.6 to 100.0) 64.1 (47.9 to 77.6) 96.0 (92.4 to 100.0)
2 years 92.7 (88.3 to 100.0) 54.5 (45.1 to 63.7) 99.0 (90.4 to 100.0) 7 B: for adolescents and adults-n= 273	87.6 (77.5 to 93.5) 94.1 (71.2 to 100.0) 90.4 (69.1 to 97.5) 72.6 (53.6 to 100.0)
B: for adolescents and adults-n= 273 CCA Sensitivity Sensitivity Common Specificity Sensitivity Common Specificity Common Sensitivity Common Sensity Common Sensitivity Common Sen	71.5 (53.4 to 84.6) 93.8 (88.6 to 100.0) 64.1 (47.9 to 77.6) 96.0 (92.4 to 100.0)
CCA KK (2 measurements) Sensitivity Specificity	
Sensitivity Specificity Sensitivity	rts) 1 st day KK (2 measurements) 2 nd day KK (2 measurement
	Specificity Specificity Sensitivity
Baseline 90.1 (84.3 to 100.0) 73.6 (68.0 to 78.6) 94.0 (89.5 to 100.0) 8	81.5 (74.5 to 87.0) 92.7 (87.9 to 100.0) 85.8 (78.7 to 90.8) 95.4 (90.2 to 100.0)
9 weeks 90.1 (84.3 to 100.0) 73.6 (68.0 to 78.6) 49.7 (35.8 to 100.0) 5	95.6 (90.0 to 98.2) 60.0 (41.9 to 100.0) 98.7 (93.7 to 99.8) 54.1 (38.1 to 100.0)
2 years 90.1 (84.3 to 100.0) 73.6 (68.0 to 78.6) 94.0 (89.5 to 100.0) 8	81.5 (74.5 to 87.0) 92.7 (87.9 to 100.0) 85.8 (78.7 to 90.8) 95.4 (90.2 to 100.0)

Latent Markov Models and Diagnostics

10 000 ng/ml spiked QC samples. Strips were scored in a blinded fashion by at least two individuals and in case of discrepancies a third person was consulted to conclude on the score. The classification of the trace result is decided later (see 'Model selection' section) based on the use of specific latent variable models. All other positive results (scores 1, 2 and 3) were merged into one positive category.

Latent Markov Models and Their Application to Evaluating Diagnostics for *S. mansoni* Infection

LMM consists of a *structural model* for the latent infection states (analogous to the latent classes in Latent Class Analysis) and a *measurement model* for the observed indicators (these are the four diagnostic tools: the average of two KK measurements on three consecutive days and the CCA), conditional on latent infection state.

Let $\mathbf{Y} = (y_{i1b}, \dots, y_{iPl})$ be a response pattern for the *i*th individual at time *t* on *P* observed binary indicators (in this study *P*=4 binary diagnostic tests) with values ' θ ' and '*1*' indicating negative and positive diagnostic test results, respectively.

We assume that for each individual the true underlying infection state, at each discrete time point t (where t=1,...T and T=3: baseline, nine weeks and two years) is explained by a latent categorical variable denoted by C with two latent infection states (i.e. those with *S. mansoni* infection and those without).

The responses to the (P×T) y indicators are assumed to be independent conditional on the latent infection state membership which in our analysis implies that the results from the four diagnostics are assumed to be independent conditional on the true underlying infection state both within and across time points. In addition, the latent categorical variable $C_{i,t}$ depends on $C_{i,t-1}$ but not on earlier latent categorical variables, known as the *first-order Markov* property. Under these assumptions, the probability of observing a particular response pattern **Y** for a randomly selected individual i is:

$$P(\mathbf{Y}_{i} = \mathbf{y}_{i}) = \sum_{j_{1}=1}^{2} \dots$$

$$\sum_{j_{T_{i}=1}}^{2} P(C_{i,1} = j_{1}) \prod_{t=2}^{T_{i}} P(C_{i,t} = j_{t}|C_{i,t-1} = j_{t-1}) \prod_{t=1}^{T_{i}} \prod_{p=1}^{p} \rho_{ptj_{t}}^{(1-y_{ipt})} (1 - \rho_{ptj_{t}})^{y_{ipt}}$$

where $P(C_{i,1} = j_1)$ represents the baseline infection state prevalence at the first time point (i.e. baseline), and $P(C_{i,t} = j_t | C_{i,t-1} = j_{t-1})$ represents the transition probability to latent infection state j_t at time t conditional on membership in latent infection state j_{t-1} at time t-1. $\rho_{pt|j_t}$ represents the diagnostic specificity for infection when the probability of the p diagnostic test is negative conditioned on $C_{i,t}$ representing the 'Not Infected' latent state. Similarly, the diagnostic sensitivity for infection of each of the four diagnostics is obtained when the probability of the p diagnostic test is positive conditioned on $C_{i,t}$ representing the 'Infected' latent state.

Figure 1 presents our LMM in a path diagram.

Computation

We fitted several LMMs using MPLUS v. 6.1 (Muthén & Muthén, Los Angeles) [31] with full information maximum likelihood in which we assumed that missing data were missing at random and making maximum use of data from individuals with incomplete data at the time points under study, for two different age groups: a) n = 167 children of age 7–16 years old and b) n = 273 adolescents and adults of age 17–76 years old. It is well

Table 3. LMM estimated transition probability matrices.

A: for children	-n = 167						
Treatment Inte	erval						
Baseline-to-9 v	weeks						
		9 weeks					
		1 PZQ dose			2 PZQ doses		
		I-	l+		I-	l+	
Baseline	I-	1.000 ^a	0.000	I-	1.000 ^b	0.000	
	I+	0.458	0.542	I+	0.717	0.283	
Non-Treatmen	t Interval						
9 weeks-to-2 y	/ears						
		2 years					
		1 PZQ dose	1 PZQ dose		2 PZQ doses		
		I-	I+		I-	I+	
9 weeks	I-	0.437	0.563	I-	0.422	0.578	
	I+	0.202	0.798	I+	0.000	1.000 ^c	
B: for adolesce	ents and adults-n=	= 273					
Treatment Inte	erval						
Baseline-to-9 v	weeks						
		9 weeks					
		1 PZQ dose	1 PZQ dose		2 PZQ doses		
		I-	l+		I-	l+	
Baseline	I-	1.000 ^a	0.000	I-	1.000 ^b	0.000	
	l+	0.561	0.439	I+	0.846	0.154	
Non-Treatmen	t Interval						
9 weeks-to-2 y	/ears						
		2 years					
		1 PZQ dose			2 PZQ doses		
		I-	l+		I-	I+	
9 weeks	I-	0.858	0.142	I-	0.836	0.164	
	I+	0.126	0.874	I+	0.000	1.000 ^c	

a, b, CThese parameters were estimated close to 1 or 0 and so in order to avoid numerical instability in the estimation algorithm, MPLUS fixed these automatically to 1 or 0 respectively. Such values should be treated with caution due to computational limitations in these categories during the model estimation.

General note: The displayed results are derived by exploiting in the final LMM the interrelationships between CCA and double Kato-Katz (KK) faecal slides over 3 consecutive days as well as not assuming a gold standard.

doi:10.1371/journal.pcbi.1003402.t003

known that different contact patterns with infected water bodies and consequently acquired exposure, immunity and susceptibility to infection might be experienced by different age groups [32].

3) The 'true' prevalence of *S. mansoni* infection at each time point.

We thus estimated the following sets of parameters for these two age groups:

Model Selection and Classification of a "Trace" CCA Test

1) Probabilities of positive or negative responses from KK and CCA diagnostic tests conditional on the latent 'infection' state at each time point (i.e. diagnostic sensitivities and specificities); in latent variable modelling literature these are called *item response probabilities*.

Transition probabilities of the latent infection states between 2)the study time points i.e. from baseline to nine weeks and from nine weeks to two years.

Line

We selected the LMMs that optimally combined goodness of fit and parsimony as measured using the Akaike Information Criterion (AIC) and the Bayesian Information Criterion (BIC). [33] Both AIC and BIC are penalized-likelihood criteria. AIC is an estimate of a constant plus the relative distance between the unknown true likelihood function of the data and the fitted likelihood function of the model, so that a lower AIC means a model is considered to be better. BIC is an estimate of a function



Figure 1. Latent Markov modelling path diagram. The variables in boxes represent the four observed categorical indicators of the latent categorical variables *C* at each time point *t*. The three arrows between the circled variables indicate the regression model for the latent categorical variable at time point *t* on the latent categorical variable at time point *t*-1. doi:10.1371/journal.pcbi.1003402.g001

of the posterior probability of a model being true, under a particular Bayesian framework, so that a lower BIC means that a model is considered to be better. BIC penalizes model complexity more heavily and thus, whenever there is an inconsistency AIC indicates a preference for a more complex model than BIC [34]. Both criteria are based on assumptions and asymptotic approximations. Each, despite its heuristic usefulness, has therefore been criticized as having questionable validity for real-world data [34,35]. In the current study, we examined AIC and BIC as well sample-size-adjusted BIC among models that were considered biologically plausible for the epidemiological settings under study (given the effect of one and two PZQ doses on schistosomiasis cure rates within the studied age groups).

We first tested whether the interpretation of a 'trace' CCA test line should be classified as positive or negative as several studies have reported on the ambiguity of infection status among those classified as 'trace' [16,17,36,37]. For this problem we used a one factor analysis model [38] on the baseline data, to explore the interrelationships among the four diagnostic tests for both age groups, treating the three results of CCA ('negative', 'trace' and 'positive') as nominal (unordered). We plotted the posterior distribution of the latent variable given the three possible responses to CCA for children (Figure 2) and adolescents and adults (Figure 3), after estimating the one factor analysis model. Having observed the similarities in the posterior distributions for the 'trace' and 'negative' categories in Figures 2 and 3, we decided to treat 'negative' and 'trace' CCA as a single 'negative' category for the rest of this analysis. Similar statistical analysis might be useful for data related to point of care-CCA in order to further validate results and proceed with generalization of recommendations about trace results for the CCA in the field.

We subsequently fitted LMMs where we initially tested whether the item response probabilities (ρ 's, i.e. the sensitivities and specificities) assumed to be the same at baseline and two years but allowed to vary at nine weeks for both KK and CCA tests. In other words we tested the hypothesis of measurement invariance which assumes the equality of the parameters of the measurement model i.e. the conditional item response probabilities for the latent infection states at the different time points. We did not test for different ρ 's at each of the three time points (baseline, nine weeks and two years) because if they did differ at each time point the meaning of the latent infection states would have changed over time making the transition probabilities τ 's uninterpretable. The reason for this is that along with interpreting quantitative change



Figure 2. Posterior distributions of latent variables conditioned on the responses of the CCA test for children (n = 167). The latent variable represents a continuum of the infection severity characterized from the left to the right as to low or no infection up to high infection. The obtained scores in the horizontal axis are linked to ranking and not necessarily to the absolute displayed values. doi:10.1371/journal.pcbi.1003402.g002



Figure 3. Posterior distributions of latent variables conditioned on the responses of the CCA test for adolescents and adults (n = 273). The latent variable represents a continuum of the infection severity characterized from the left to the right as to low or no infection up to high infection. The obtained scores in the horizontal axis are linked to ranking and not necessarily to the absolute displayed values. doi:10.1371/journal.pcbi.1003402.g003

over time in latent infection state membership (i.e. through the τ 's), it also becomes necessary to interpret change over time in the meaning of the latent infection state (i.e. which latent infection state represents the 'true' negatives and 'true' positives at each time point).

We also tested whether the provision of one or two PZQ doses affected the transition probabilities τ 's in the non-treatment intervals (nine weeks and two years) or only in the treatment interval (baseline to nine weeks) since there might be long-term benefits of two PZQ doses in the absence of snail control as in this community. Equal transition probability τ 's across each of the treatment intervals under study was not tested because this was not consistent with prior knowledge of the MDA impacts.

Although there was information identifying the household of each individual in the study, there were not enough data within each of the age groups studied to take into account betweenhousehold variability in the considered models.

Results

The observed numbers of "true" and "false positives" and "true" and "false negatives", and corresponding observed sensitivity and specificity, for the CCA diagnostic test are presented assuming, for illustration, the combination of three duplicate KK measures over the three consecutive days as being both 100% sensitive and 100% specific (Table 1).

When the sensitivities and specificities of single-day double KK results were estimated by fitting LMMs, the information criteria indicated that, for both age groups, the transition probability matrices [for the baseline-to-nine weeks follow-up (treatment interval) and for the nine weeks-to-two years follow-up (non-treatment interval)] depended on the provision of one or two PZQ doses. Similarly, the measurement invariance hypothesis for CCA was accepted based on the information criteria; however, the measurement invariance hypothesis was rejected for KK (i.e. the diagnostic performance of KK was found to vary over time but that of CCA was not). For further details on final model selection, see Table S1. The estimated sensitivities and specificities for CCA and KK tests, from the best LMM for each age group are presented in Table 2. Based on these estimates and if one took the crude average of the estimated sensitivities and specificities of the two diagnostic

measures and time points under examination, CCA overall was found to be more sensitive but less specific than double KK from a single faecal sample for both age groups. The sensitivity of double KK from a single faecal sample was found to be lower at nine weeks than at baseline and two years for both age groups in all three days for adolescents and adults and for two of the three days for children.

The estimated transition probabilities between the latent infection states over time are presented in Table 3. Diagonal elements represent the probabilities of remaining in the same latent infection state at time t as at the previous time (t-1). The majority of adolescents and adults apparently remained, over the non-treatment interval, in the same latent infection state. Among those infected at baseline in both age groups, two PZQ doses produced a higher probability of clearance of infection at nine weeks than just one PZQ dose; for children these probabilities were 0.717 (2 doses) and 0.458 (1 dose) (see Table 3A), while for adolescents and adults these probabilities were 0.846 (2 doses) and 0.561 (1 dose) (see Table 3B).

The *S. mansoni* prevalence estimate for each time point and each age group is presented in Figure 4 with 95% confidence intervals. Following treatment the prevalence of *S. mansoni* infection decreased dramatically between baseline to nine weeks for both age groups. Among children there is a substantial rebound by two years. These patterns are also reflected in the transition probabilities in Table 3.

Although visual examination of Table 4 superficially suggests that the LMM estimate conflicts with that obtained based on the 6 KK measurements (assumed positive if one or more of the six measurements was positive), further calculations are required to underpin full interpretation of the results.

For instance, for children at *baseline*, the estimated prevalence based on the LMM is 92.6%.

One would expect the following percentage of **true** positives:

$$92.6\% * (1-(1-.990)*(1-.938)*(1-.96)) = 92.6\% * 0.999975 = 92.6\%$$

where 0.990, 0.938 and 0.960 are the estimated sensitivities of two KK measurements from days one, two and three, respectively



Figure 4. LMM estimated "true" S. mansoni prevalence for both age groups (ages 7–16 and 17–76) with 95% confidence intervals over time.

doi:10.1371/journal.pcbi.1003402.g004

One would expect the following percentage of **false** positives:

$$(100\% - 92.6\%) * (1 - .715 * .641 * .709)$$

= 7.4\% * 0.675005 = 5.0\%

where 0.751, 0.641 and 0.709 are the estimated specificities of two KK measurements from days one, two and three, respectively.

Thus, based on the estimated sensitivities, specificities and the assumption that results from days one, two and three, were independent, conditional on the true status, the LMM predicts that the estimated prevalence based on 6 KK measurements would be **97.6%** (92.6% true positive and 5.0% false positives). This is highly consistent with the estimate obtained from the 6 KK measurements: **97.6%**.

For the remaining time points and age groups please see Text S2 in Supporting information.

Discussion

This study analysed a longitudinal detailed dataset from Uganda in which one or two doses of PZQ treatment were provided at baseline using LMM that accounted for the longitudinal study design and the measurement error in the diagnostic methods under study. Our primary objective was to

Table 4. Estimated S. mansoni prevalences	based	on	6	KK
measurements and based on the LMM.				

Source of the estimate	Baseline	9 weeks	2 years
	Children		
6 KK measurements	97.6	62.1	90.8
Latent Markov model	92.6	38.0	67.9
	Adolescents and Adults		
6 KK measurements	83.9	30.0	56.5
Latent Markov model	72.9	22.5	31.3

doi:10.1371/journal.pcbi.1003402.t004

assess the CCA diagnostic accuracy at baseline and at two followup times after treatment but we also evaluate double KK faecal slides over three consecutive days for *S. mansoni* infection. To our knowledge, this is the first study which provides rigorous modelbased diagnostic performance of CCA and single-day double KK measurements over three consecutive days for the diagnosis of *S. mansoni* infection in two different age groups pre- and post- PZQ treatment.

CCA's diagnostic performance was found to be constant over time and overall approximately 90% sensitive but less specific than single-day double KK faecal slides for *S. mansoni* infection in both age groups. Day-to-day variation in faecal egg output among *Schistosoma* parasites [7–9] has been shown to be greater [11] and with lower sensitivity of KK after PZQ treatment [8]. The singleday double KK sensitivity is likely to depend strongly on the observed prevalence [8,15,39]. Our study confirmed these findings and arguments and showed clearly that sensitivity of single-day double KK was much lower at nine weeks than at baseline and two years for both age groups in all three days for adolescents and adults and for two of the three days for children while its specificity increased after PZQ treatment (Table 2).

These findings bridge existing gaps in schistosomiasis diagnostics research such as for instance the lack of CCA evaluation in adolescents and adults and the lack for evidence for its capacity to determine if a person has been cured after treatment, as previously highlighted [4,17]. The current analysis provides model-based estimates of sensitivity and specificity and their uncertainties (through the provision of 95% confidence intervals) without assuming any gold standard diagnostic test in the statistical analysis. The exact numbers of false positive and false negative results are almost always unknown and thus in the current study we estimated rather than assumed values for the parameters displayed in Tables 2 and 3 (sensitivities, specificities and transition probabilities) [40,41]. Without quantification of the uncertainties regarding the performance of the key diagnostic tests, generalization of epidemiological results and development of useful recommendations for which diagnostics to use and at which stages of schistosomiasis control are hampered.

This approach evaluates the risk of potential misinterpretation with regards to diagnosis of *S. mansoni* infection through CCA or KK in this endemic setting pre- and post- PZQ treatment as the numbers and infection intensities are brought down [12]. For instance, results in Table 1 demonstrate that by using 6 KK measurements over three days as the gold standard (i.e. assuming 100% sensitivity and 100% specificity), the resulting empirical estimates of CCA sensitivity and specificity are mistakenly shown to vary over time. We do not expect that the clearance of the antigen is influenced by treatment. Furthermore, the hypothesis of measurement invariance was not rejected based on information criteria for the fitted LMMs (see Text S1 and Table S1 in Supporting Information).

Glinz et al. 2010 [42] discussed possible reasons for false positive diagnoses from KK tests. Results from model in Table 2 clearly indicate that the specificity of single-day double KK measurements is lower than 100%. This means that despite highly qualified and skilled co-workers, contamination of stool sieves in the field and data entry errors cannot completely be avoided. Because our estimated specificities of single-day double KK measurements were less than 100%, the estimated 'true' S. mansoni prevalence (Figure 4) is lower at each time points than the estimated prevalence obtained assuming that any individual with positive results on one or more of the six KK tests conducted over three consecutive days was infected (Tables 1 and 4). This is in accordance with work on diagnostic performance of KK for animal schistosomiasis infection [14]. Previous work using stochastic models have demonstrated that the sensitivity of the KK would vary according to the number of stool samples provided [7-9] and the 'true' S. mansoni prevalence at baseline can be calculated using the De Vlas pocket chart [8]. For the children group, the chart is not applicable since the observed prevalence in this study is beyond the limits where the De Vlas model is valid and thus we cannot compare it with the estimates of 'true' prevalence from our model (Figure 4). For the adolescents and adults group however estimates of the 'true' prevalence (Figure 4) were not consistent with this chart because it was based on an assumption of 100% KK specificity.

The estimated transition probability matrices (Table 3) provide parsimonious yet important insights into the re-infection and cure rates in the two age groups. The cure rate was higher in adolescents and adults than in children following treatment. This can be explained by the fact that those infected in the older age group had lower burdens than the infected children and would be therefore more likely to become negative. From an immunological perspective view, it can be argued that the older age group are more likely to have developed protective immune response and are therefore more efficient in affecting and killing the worms [43,44]. The quantification of the additional benefit of a second PZQ dose in reducing infection levels for both age groups was demonstrated by higher transition probabilities from infected to non-infected among those who received two PZQ doses compared to those who received one PZQ dose within the treatment interval. For the non treatment interval (i.e. between nine weeks and two years) there were no differences in reinfection or cure rates among those who received one or two PZQ doses. As the LMMs are estimated using an iterative algorithm, at each step of the algorithm, estimated values very close to 0 or 1 can create estimation instability and therefore are automatically from MPLUS fixed to 0 and 1 respectively in order to avoid non-convergence of the estimation algorithm. Consequently, such values should be treated with caution due to computational limitations in these categories during the model estimation. Finally, we recognize that the results of this study depend upon the assumption of conditional independence assumed by the models fitted here. Once one conditions on the latent infection state, we believe though that there are good reasons to assume that KK and CCA are independent as *Schistosoma* eggs and antigens are excreted through different routes in the human body, for instance.

To conclude, in the absence of a diagnostic gold standard this study has demonstrated that LMMs can be useful for the evaluation of available diagnostic tools for S. mansoni infection. More generally, we recommend LMMs to be used for the evaluation of diagnostic tests of other diseases without gold standard diagnostic tools whenever longitudinal data are available as such modelling permits questions about changes in true infection states and test the measurement invariance hypothesis of the diagnostic tests of interest over time - making them very useful tools indeed for control program M&E research [21]. Further work in evaluating the trace result and the ability of CCA to quantify intensity of infection is also warranted in the M&E of schistosomiasis control programs and dynamic latent factor models (such models would assume continuous hypothetical constructs or typologies-i.e. intensity of infection) might be appropriate statistical methods for the analysis of relevant data. Similar studies should be considered at other sites in order to build on our results. We found that the CCA diagnostic performance remained constant after provision of PZQ treatment and that the test is overall more sensitive but less specific than single-day double KK for the diagnosis of S. mansoni infection. In line with the results from our study and those of a recent multi-country cross-sectional study which showed that for lower S. mansoni intensity settings the CCA sensitivity was demonstrated to be higher than KK [17], we recommend that CCA to be used for mapping surveys of S. mansoni infection. As public health measures are aimed at the elimination of residual foci of schistosomes, data generated using diagnostics with high specificity will be required to avoid unnecessarily prolonging MDA and wasting scarce resources [5]. Detection of parasite-specific DNA [45,46] or circulating anodic antigen in serum or urine [47] might present alternative opportunities in schistosomiasis elimination programs and further evaluations of these diagnostics merit attention.

Supporting Information

Table S1Different fitted LMMs.(DOCX)

Text S1 Comparison of Alternative Latent Structures of LMMs. (DOCX)

Text S2 Validation of LMM estimates with observed 6 KK measurements.

(DOCX)

Text S3 Comparisons of estimated *S. mansoni* prevalences with the De Vlas pocket chart. (DOCX)

Acknowledgments

We would like to acknowledge the willing help and collaboration of individuals who participated in this study. We are also grateful to Prof Melanie Wall for her advice in some of our MPLUS codes.

Author Contributions

Conceived and designed the experiments: BJV EMT NBK SW. Performed the experiments: EMT NBK SW. Analyzed the data: AK. Contributed reagents/materials/analysis tools: GJvD. Wrote the paper: AK BJV GJvD. Conceived and designed the field study: BJV EMT NBK SW. Performed the field sample and data collection: EMT NBK SW. Performed the CCA testing: GJvD. Conceived and conducted the statistical analysis: AK. Assisted and advised with complex statistical modelling: CAD IM. Developed the computer code: AK. Read and commented critically on the manuscript: CAD IM EMT NBK SW JPW AMD.

References

- Fenwick A, Webster JP (2006) Schistosomiasis: challenges for control, treatment and drug resistance. Curr Opin Infect Dis 19: 577–582.
- McCarthy JS, Lustigman S, Yang GJ, Barakat RM, Garcia HH, et al. (2012) A research agenda for helminth diseases of humans: diagnostics for control and elimination programmes. PLoS Negl Trop Dis 6: e1601.
- Vercruysse J, Behnke JM, Albonico M, Ame SM, Angebault C, et al. (2011) Assessment of the anthelmintic efficacy of albendazole in school children in seven countries where soil-transmitted helminths are endemic. PLoS Negl Trop Dis 5: e948.
- Shiff C (2012) The Importance of Definitive Diagnosis in Chronic Schistosomiasis, with Reference to Schistosoma haematobium. J Parasitol Res 2012: 761269.
- Solomon AW, Engels D, Bailey RL, Blake IM, Brooker S, et al. (2012) A diagnostics platform for the integrated mapping, monitoring, and surveillance of neglected tropical diseases: rationale and target product profiles. PLoS Negl Trop Dis 6: e1746.
- Utzinger J, N'Goran E K, Caffrey CR, Keiser J (2011) From innovation to application: social-ecological context, diagnostics, drugs and integrated control of schistosomiasis. Acta Trop 120 Suppl 1: S121–137.
- De Vlas SJ (1996) Modelling human Schistosoma mansoni infection: the art of counting eggs in faeces. [PhD Thesis]. Rotterdam.
- De Vlas SJ, Gryseels B (1992) Underestimation of Schistosoma mansoni prevalences. Parasitol Today 8: 274–277.
- De Vlas SJ, Gryseels B, van Oortmassen GJ, Polderman AM, Habbema JD (1993) A pocket chart to estimate true *Schistosoma mansoni* prevalences. Parasitol Today 9: 305–307.
- Engels D, Sinzinkayo E, De Vlas SJ, Gryseels B (1997) Intraspecimen fecal egg count variation in *Schistosoma mansoni* infection. Am J Trop Med Hyg 57: 571–577.
- Utzinger J, Booth M, N'Goran EK, Muller I, Tanner M, et al. (2001) Relative contribution of day-to-day and intra-specimen variation in faecal egg counts of *Schistosoma mansoni* before and after treatment with praziquantel. Parasitology 122: 537–544.
- Bergquist R, Johansen MV, Utzinger J (2009) Diagnostic dilemmas in helminthology: what tools to use and when? Trends Parasitol 25: 151–156.
- Booth M, Vounatsou P, N'Goran E K, Tanner M, Utzinger J (2003) The influence of sampling effort and the performance of the Kato-Katz technique in diagnosing *Schistosoma mansoni* and hookworm co-infections in rural Cote d'Ivoire. Parasitology 127: 525–531.
- Carabin H, Balolong E, Joseph L, McGarvey ST, Johansen MV, et al. (2005) Estimating sensitivity and specificity of a faecal examination method for *Schistosoma japonicum* infection in cats, dogs, water buffaloes, pigs and rats in Western Samar and Sorsogon Provinces, The Philippines. Int J Paras 35: 1517– 1524.
- Coulibaly JT, Knopp S, N'Guessan NA, Silue KD, Furst T, et al. (2011) Accuracy of urine circulating cathodic antigen (CCA) test for *Schistosoma mansoni* diagnosis in different settings of Cote d'Ivoire. PLoS Negl Trop Dis 5: e1384.
- Tchuem Tchuente LA, Kuete Fouodo CJ, Kamwa Ngassam RI, Sumo L, Dongmo Noumedem C, et al. (2012) Evaluation of circulating cathodic antigen (CCA) urine-tests for diagnosis of *Schistosoma mansoni* infection in Cameroon. PLoS Negl Trop Dis 6: e1758.
- Colley DG, Binder S, Campbell C, King CH, Tchuem Tchuente LA, et al. (2013) A Five-Country Evaluation of a Point-of-Care Circulating Cathodic Antigen Urine Assay for the Prevalence of *Schistosoma mansoni*. Am J Trop Med Hyg 88: 426–432.
- Gutman J, Fagbemi A, Alphonsus K, Eigege A, Miri ES, et al. (2008) Missed treatment opportunities for schistosomiasis mansoni, in an active programme for the treatment of urinary schistosomiasis in Plateau and Nasarawa states, Nigeria. Ann Trop Med Parasitol 102: 335–346.
- van Dam GJ, Wichers JH, Ferreira TM, Ghati D, van Amerongen A, et al. (2004) Diagnosis of schistosomiasis by reagent strip test for detection of circulating cathodic antigen. J Clin Microbiol 42: 5458–5461.
- Sousa-Figueiredo JC, Betson M, Atuhaire A, Arinaitwe M, Navaratnam AM, et al. (2012) Performance and safety of praziquantel for treatment of intestinal schistosomiasis in infants and preschool children. PLoS Negl Trop Dis 6: e1864.
- Koukounari A, Moustaki I, Grassly NC, Blake IM, Basáñez MG, et al. (2013) Using a Nonparametric Multilevel Latent Markov Model to Evaluate Diagnostics for Trachoma. Am J Epidem 2013 177: 913–922.
- Langcheine R, van de Pol F (1990) A Unifying Framework for Markov Modeling in Discrete Space and Discrete Time. Sociol Method Res 18: 416–441.
- Tukahebwa ÉM, Magnussen P, Madsen H, Kabatereine NB, Nuwaha F, et al. (2013) A very high infection intensity of *Schistosoma mansoni* in a Ugandan Lake Victoria fishing community is required for association with highly prevalent organ related morbidity. PLoS Negl Trop Dis 7: e2268. doi: 10.1371/ journal.pntd.0002268.
- Kabatereine NB, Brooker S, Koukounari A, Kazibwe F, Tukahebwa EM, et al. (2007) Impact of a national helminth control programme on infection and morbidity in Ugandan schoolchildren. Bull World Health Organ 85: 91–99.

- Koukounari A, Fenwick A, Whawell S, Kabatereine NB, Kazibwe F, et al. (2006) Morbidity indicators of *Schistosoma mansoni*: relationship between infection and anemia in Ugandan schoolchildren before and after praziquantel and albendazole chemotherapy. Am J Trop Med Hyg 75: 278–286.
- Tukahebwa EM, Vennervald BJ, Nuwahab F, Kabatereine NB, Magnussen P (2013) Comparative efficacy of one versus two doses of praziquantel on cure rate of *Schistosoma mansoni* infection and re-infection in Mayuge District, Uganda. Trans R Soc Trop Med Hyg 107: 397–404.
- Koroma JB, Peterson J, Gbakima AA, Nylander FE, Sahr F, et al. (2010) Geographical distribution of intestinal schistosomiasis and soil-transmitted helminthiasis and preventive chemotherapy strategies in Sierra Leone. PLoS Negl Trop Dis 4: e891.
- Legesse M, Erko B (2007) Field-based evaluation of a reagent strip test for diagnosis of *Schistosoma mansoni* by detecting circulating cathodic antigen in urine before and after chemotherapy. Trans R Soc Trop Med Hyg 101: 668—673.
- Odiere MR, Rawago FO, Ombok M, Secor WE, Karanja DM, et al. (2012) High prevalence of schistosomiasis in Mbita and its adjacent islands of Lake Victoria, western Kenya. Parasit Vectors 5: 278.
- Midzi N, Butterworth AE, Mduluza T, Munyati S, Deelder AM, et al. (2009) Use of circulating cathodic antigen strips for the diagnosis of urinary schistosomiasis. Trans R Soc Trop Med Hyg 103: 45–51.
- Muthén LK, Muthén BO (2010) Mplus Statistical Analysis with Latent Variables, User's Guide. Los Angeles, CA: Muthén & Muthén.
- Van Lieshout L, Polman K, Gryseels B, Deelder AM (1998) Circulating anodic antigen levels in two areas endemic for schistosomiasis mansoni indicate differences in worm fecundity. Trans R Soc Trop Med Hyg 92: 115–119.
- Burnham K P., Anderson D R (2004) Multimodel inference: Understanding AIC and BIC in model selection. Sociological Methods and Research 33: 261– 304.
- Dziak JJ, Coffman DL, Lanza ST, Li R (2012) Sensitivity and specificity of information criteria. The Methodology Center and Department of Statistics, Penn State, The Pennsylvania State University.
- Bozdogan H (1987) Model-selection and Akaike's information criterion (AIC): The general theory and its analytical extensions. Psychometrika 52: 345–370.
- Shane HL, Verani JR, Abudho B, Montgomery SP, Blackstock AJ, et al. (2011) Evaluation of urine CCA assays for detection of *Schistosoma mansoni* infection in Western Kenya. PLoS Negl Trop Dis 5: e951.
- Stothard JR, Sousa-Figuereido JC, Betson M, Adriko M, Arinaitwe M, et al. (2011) Schistosoma mansoni Infections in young children: when are schistosome antigens in urine, eggs in stool and antibodies to eggs first detectable? PLoS Negl Trop Dis 5: e938.
- Bartholomew DJ, Knott M, Moustaki I (2011) Polytomous data: latent trait models. Latent Variable Models and Factor Analysis: A Unified Approach. 3rd ed.
- Berhea N, Medhina G, Erkoa B, Smith T, Gedamu S, et al. (2004) Variations in helminth faecal egg counts in Kato–Katz thick smears and their implications in assessing infection status with *Schistosoma mansoni*. Acta Trop. 92 205–212
- Joseph L, Gyorkos TW, Coupal L (1995) Bayesian estimation of disease prevalence and the parameters of diagnostic tests in the absence of a gold standard. Am J Epidemiol 141: 263–272.
- Tarafder MR, Carabin H, McGarvey ST, Joseph L, Balolong E, Jr., et al. (2011) Assessing the impact of misclassification error on an epidemiological association between two helminthic infections. PLoS Negl Trop Dis 5: e995.
- 42. Glinz D, Silue KD, Knopp S, Lohourignon LK, Yao KP, et al. (2010) Comparing diagnostic accuracy of Kato-Katz, Koga agar plate, etherconcentration, and FLOTAC for *Schistosoma mansoni* and soil-transmitted helminths. PLoS Negl Trop Dis 4: e754.
- 43. Kabatereine NB, Vennervald BJ, Ouma JH, Kemijumbi J, Butterworth AE, et al. (1999) Adult resistance to schistosomiasis mansoni: age-dependence of reinfection remains constant in communities with diverse exposure patterns. Parasitology 118: 101–105.
- 44. Pinot de Moira A, Fulford AJ, Kabatereine NB, Ouma JH, Booth M, et al. (2010) Analysis of complex patterns of human exposure and immunity to Schistosomiasis mansoni: the influence of age, sex, ethnicity and IgE. PLoS Negl Trop Dis 4: e820.
- Enk MJ, Oliveira ESG, Rodrigues NB (2010) A salting out and resin procedure for extracting *Schistosoma mansoni* DNA from human urine samples. BMC Res Notes 3: 115.
- 46. Ibironke O, Koukounari A, Asaolu S, Moustaki I, Shiff C (2012) Validation of a new test for *Schistosoma haematobium* based on detection of Dra1 DNA fragments in urine: evaluation through latent class analysis. PLoS Negl Trop Dis 6: e1464.
- Corstjens PL, van Lieshout L, Zuiderwijk M, Kornelis D, Tanke HJ, et al. (2008) Up-converting phosphor technology-based lateral flow assay for detection of Schistosoma circulating anodic antigen in serum. J Clin Microbiol 46: 171–176.